Author’s response to reviews

Title: Analysis of Cardiovascular Mortality, Bleeding, Vascular and Cerebrovascular Events in Patients with Atrial Fibrillation vs. Sinus Rhythm Undergoing Transfemoral Transcatheter Aortic Valve Implantation (TAVR)

Authors:

Joerg Herold (Joerg_herold@hotmail.com)
Vasiliki Herold-Vlanti (v1985@windowslive.com)
Mohammad Sherif (mohammad.sherif@med.uni-rostock.de)
Blerim Luani (blerim.Luani@med.ovgu.de)
Klaus Bonaventura (kbonaventura@klinikumevb.de)
Christin Breyer (christinbreyer@gmx.de)
Ruediger Braun-Dullaeus (r.Braun-Dullaeus@med.ovgu.de)

Version: 2 Date: 23 Sep 2017

Author’s response to reviews:

Dear Prof. Marzia Cottini

Thank you very much for considering our manuscript "Analysis of Cardiovascular Mortality, Bleeding, Vascular and Cerebrovascular Events in Patients with Atrial Fibrillation vs. Sinus Rhythm Undergoing Transfemoral Transcatheter Aortic Valve Implantation (TAVR)" (BCAR-D-17-00421R1).

We have attached the figures and tables as recommended.

Editor Comments:

Please resent the paper with figure and eventually table.
- The figures and tables have been attached.
Declaration of competing interests:

We declare that we have no competing interests to following questions below:

1. Have you in the past five years received reimbursements, fees, funding, or salary from an organisation that may in any way gain or lose financially from the publication of this manuscript, either now or in the future?

2. Do you hold any stocks or shares in an organisation that may in any way gain or lose financially from the publication of this manuscript, either now or in the future?

3. Do you hold or are you currently applying for any patents relating to the content of the manuscript?

4. Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript?

5. Do you have any other financial competing interests?

6. Do you have any non-financial competing interests in relation to this paper?

Best regards

PD. Dr. med. Joerg Herold

Magdeburg, 23.09.2017